Author’s response to reviews

Title: Do we understand the intervention? What complex intervention research can teach us for the evaluation of Clinical Ethics Support Services (CESS).

Authors:

Jan Schildmann (jan.schildmann@medizin.uni-halle.de)
Stephan Nadolny (stephan.nadolny@fh-bielefeld.de)
Joschka Haltaufderheide (joschka.haltaufderheide@rub.de)
Marjolein Gysels (mhgysels@gmail.com)
Jochen Vollmann (jochen.vollmann@rub.de)
Claudia Bausewein (Claudia.Bausewein@med.uni-muenchen.de)

Version: 2 Date: 15 May 2019

Author’s response to reviews:

We would like to thank reviewers and editors for their detailed and helpful comments. Please find our responses on the respective comments in the attached document (response to review). In addition all changes in the manuscript have been made as tracked change or highlighted.